Author's response to reviews

Title: Zinc/copper imbalance reflects immune dysfunction in human leishmaniasis: an ex vivo and in vitro study

Authors:

Johan Van Weyenbergh (johan@cpqgm.fiocruz.br)
Giselia Santana (gigisanta@svn.com.br)
Argemiro D’Oliveira Jr (argemiro@ufba.br)
Anibal F Santos Jr (bal@uefs.br)
Carlos H Costa (CrlsHNCst@aol.com)
Edgar M Carvalho (edgar@ufba.br)
Aldina Barral (abarral@cpqgm.fiocruz.br)
Manoel Barral-Netto (mbarral@cpqgm.fiocruz.br)

Version: 3 Date: 12 November 2004

Author's response to reviews:

Dear Madam,
Dear Sir,

Enclosed you may find the final version of the manuscript, with the following corrections:
Title - all unnecessary capitalisation has been removed.
Highlighting/Tracking - has been removed
Affiliations - Each affiliation written in full
References - References cited in the text using square brackets
Competing interests - sentence corrected.
References - The reference list contains authors' names
Figure titles - Full stop after each figure title.

Yours sincerely,
Johan Van Weyenbergh